Author’s response to reviews

Title: 'If nurses were in our shoes would they breastfeed their own babies?' A qualitative inquiry on challenges of breastfeeding mothers on the PMTCT programme in a rural community in Zimbabwe

Authors:

Zibusiso Nyati-Jokomo (jokomob@gmail.com)

Inam Chitsike (chitsikei@gmail.com)

Elizabeth Mbizvo (elizabethmbizvo@hotmail.com; mbizvoie@gmail.cm)

James January (jamesjanuary4@gmail.com)

Version: 2 Date: 12 Feb 2019

Author’s response to reviews:

Dear Editor,

We have responded to the issues you raised and added the interview guide we used to elicit information from the breast feeding women who participated in the study.